Author’s response to reviews

Title: Oral Status, Oral Health knowledge, Attitudes and Behavior in rural children of Shanxi, western China

Authors:

Jianghong Gao (mamaluck@sohu.com)
Jianping Ruan (ruanjp@mail.xjtu.edu.cn)
Lin Zhao (zhaol@mail.xjtu.edu.cn)
Hong Zhou (zhouhong@mail.xjtu.edu.cn)
Ruizhe Huang (huangrzh@mail.xjtu.edu.cn)
Jiangang Tian (tianjiankang@mail.xjtu.edu.cn)

Version: 3 Date: 20 June 2014

Author’s response to reviews:

Dear editor Nolasco:

Thank you very much for your reviewing our manuscript and good suggestions. Now we revised the manuscript according to your suggestions. We would like to resubmit the manuscript entitled “Oral Status, Oral Health knowledge, Attitudes and Behavior in rural children of Shanxi, western China” in the journal of BMC Oral Health.

1. Title Page. We have edited it for containing the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author, however, I am not sure whether it’s appropriate for the journal of BMC Oral Health. If having any questions, Please do not hesitate to contact us.

2. The Abstract. We have made a separate page for the abstract in the revised version.

3. We have made the main manuscript to be double spaced and the references be in [] not (). “Instruction” have been replaced with “background”. I am very grateful for your thoughtful suggestion.

4. The references have been formated in line with the requirement of BMC Oral Health according your suggestion.

5. We have carefully modified the English of the manuscript. If having any other questions, we will further revised them as possible as we can.

Thank you very much once again for your comments.

Best wishes,

Sincerely yours,